Author’s response to reviews

Title: Feedback reporting of survey data to healthcare aides

Authors:

Alison M Hutchinson (alison.hutchinson@deakin.edu.au)
Neha Batra-Garga (neha.batragarga@albertahealthservices.ca)
Lisa A Cranley (lisa.cranley@nurs.ualberta.ca)
Anne-Marie Bostrom (anne-marie.bostrom@ki.se)
Greta G Cummings (greta.cummings@ualberta.ca)
Peter G Norton (norton@ucalgary.ca)
Carole A Estabrooks (carole.estabrooks@ualberta.ca)

Version: 4 Date: 21 May 2012

Author’s response to reviews:

Thanks very much for the editorial feedback. I have accepted all change made by
the copy-editor and have added a Figure legend (after the reference list in the
manuscript).

Please note that this is one of a set of 3 manuscripts that are going to be
published together and so it shouldn’t be published until they are all ready for
publication.

Thank you again,

Alison